Author’s response to reviews

Title: Pocket depth and bleeding on probing and their associations with dental, lifestyle, socioeconomic and blood variables: A cross-sectional, multicenter feasibility study of the German National Cohort

Authors:

Heiko Zimmermann (h.zimmermann@uni-heidelberg.de)
Daniel Hagenfeld (daniel.hagenfeld@googlemail.com)
Katja Diercke (Katja.Diercke@med.uni-heidelberg.de)
Nihad El-Sayed (Nihad.ElSayed@med.uni-heidelberg.de)
Julia Fricke (julia.fricke@charite.de)
Karin Halina Greiser (h.greiser@Dkfz-Heidelberg.de)
Jan Kühnisch (jkuehn@dent.med.uni-muenchen.de)
Jakob Linseisen (j.linseisen@helmholtz-muenchen.de)
Christa Meisinger (christa.meisinger@helmholtz-muenchen.de)
Nicole Pischon (Nicole.Pischon@charite.de)
Tobias Pischon (tobias.pischon@mdc-berlin.de)
Stefanie Samietz (samietzs@uni-greifswald.de)
Marc Schmitter (Marc.Schmitter@med.uni-heidelberg.de)
Astrid Steinbrecher (Astrid.Steinbrecher@mdc-berlin.de)
Ti-Sun Kim (Ti-Sun.Kim@med.uni-heidelberg.de)
Heiko Becher (h.becher@uke.de)

Version: 6 Date: 22 December 2014

Author’s response to reviews:

Dear Editors,

we submit the revised version of the original article to be considered for publication in the journal of BMC Oral Health. We have given a point-by-point response to the editorial concerns (please see below).

We have changed the sentence regarding the study approval as recommended. This change in the paper is highlighted in red.

Thank you for the possibility to resubmit to BMC Oral Health and considering our manuscript.

We look forward to hearing from you soon.

Sincerely,

Heiko Zimmermann and Heiko Becher on behalf of the authors
Dear Mr Zimmermann,

Thank you for your revised manuscript, however we require some further revisions, please amend the following statement:

The ethical approvals of the study protocols were given (Charité - Universitätsmedizin Berlin (EA1/101/11), Medizinische Fakultät der Universität Heidelberg (S-108/2011), Medizinische Fakultät der Universität Greifswald (BB 12/11), Bayerische Landesärztekammer, München).

to

The study approval was approved by the following ethics committees;

The study approval was approved by the following ethics committees; (Charité – Universitätsmedizin Berlin (EA1/101/11), Medizinische Fakultät der Universität Heidelberg (S-108/2011), Medizinische Fakultät der Universität Greifswald (BB 12/11), Bayerische Landesärztekammer, München).